Author’s response to reviews

Title: Concurrent presentation of thyroid storm and diabetic ketoacidosis: a systematic review of previously reported cases

Authors:

Devarajan Rathish (rathishdeva@gmail.com)
Senuri Karalliyadda (senuri_k@yahoo.com)

Version: 2 Date: 19 Apr 2019

Author’s response to reviews:

18th of April 2019

The Editor
BMC Endocrine Disorders

Dear Sir / Madam,

BEND-D-19-00045R1

Submission of revised manuscript and replies to reviewers’ comments

Concurrent presentation of thyroid storm with diabetic ketoacidosis: a systematic review of previously reported cases

On behalf of the authors, I would like to submit our replies to the comments by the editor along with the edited version of the manuscript. We value the comments of the editor which have helped us to improve our manuscript. Comments of the editor were accepted and necessary alterations were made to the manuscript.

We feel that the findings described in this manuscript will be of special interest to the readers of the BMC Endocrine Disorders. All authors have read the final manuscript and have approved this submission. We do not report any conflicts of interest.
Yours Sincerely,

Dr D Rathish
Lecturer in Pharmacology,
Faculty of Medicine and Allied sciences,
Rajarata University of Sri Lanka,
Saliyapura, Anuradhapura,
Sri Lanka.
Email: rathishdeva@gmail.com
Phone number: +94774771214, +94710575578

Replies to editor’s comments

Editor Comment 1
Please relocate your Additional Files list so that it is presented after the References.
Authors Response
The list of additional files was relocated after the references.

Editor Comment 2
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Authors Response
All highlights were omitted from the manuscript. Figures and additional files were uploaded as separate files.
Reviewer 1 comment 3

As this will expectedly be the final editorial revision before acceptance for publication, please take this opportunity to perform an in-depth copy edit for publication level English grammar and syntax, as it has been suggested by editor and reviewer both that a copy edit is in order prior to publication.

Authors Response

The authors have re-visited the entire manuscript for a copy edit.